Author’s response to reviews

Title: Prevalence, genetic variants and clinical implications of G-6-PD deficiency in Burkina Faso: a systematic review.

Authors:
Abdoul Karim OUATTARA (ak_ouattara@yahoo.fr)
Pouiré YAMEOGO (pouireyameogo@gmail.com)
Lassina TRAORE (ttl.lass@yahoo.fr)
Birama DIARRA (diarra.birama679@gmail.com)
Maléki ASSIH (assihmalki@yahoo.fr)
Tegwindé COMPAORE (rebecca23fr@yahoo.fr)
Dorcas OBIRI-YEBOAH (castella.oy@gmail.com)
Serge SOUBEIGA (s.soubeiga@yahoo.fr)
Florence DJIGMA (florencia.djigma@gmail.com)
Jacques SIMPORE (jacques.simpore@yahoo.fr)

Version: 3 Date: 07 Nov 2017

Author’s response to reviews:

Answer to reviewer

NB: The different corrections made as follows “section, line, page”.

1. Please include a Methods section in your Abstract.

Methods section has been included in the abstract. Abstract section, Line 33 to 37, Page 2.

2. Please check that all authors’ email addresses are the same in Editorial Manager as those on the title page, the email address for author Florencia Wendkuuni DJIGMA appears to be different.

The Email has been corrected in the manuscript: florencia.djigma@gmail.com. Title page, Line: 11, Page 1.
3. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

It is now clear state that all authors have read and approved the paper. Authors’ contributions section: Line 286, Page 14.

4. Please represent authors’ names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

The remarks were taken into account in the “authors’ contributions” section. Authors’ contributions section, Line 277 to 285, Pages 13 and 14.

5. Please clarify whether the map depicted in figure 2 is your own or taken from another source.

The map depicted in figure 2 is my own and the source has been clearly stated in the legend of Figure 2 and on the map. Figure 2 Legend, Line 484, Page 24.